Author’s response to reviews

Title: Portal vein thrombosis after aortic valve replacement surgery in a patient with antithrombin III deficiency - Case report and review of literature

Authors:

Yu-qing Wang (wyq19871001@126.com)
Da Zhu (250794259@qq.com)
Qiu-ling Chen (johnnychenhi@qq.com)
Li Dong (donglikn199@163.com)

Version: 2
Date: 19 February 2014

Author’s response to reviews: see over
Dear Editor,

We would like to submit our paper "Portal vein thrombosis after aortic valve replacement surgery in a patient with antithrombin III deficiency - Case report and review of literature". In this manuscript, we presented an unique case of PVT after aortic valve replacement due to antithrombin III deficiency. PVT after AVR is a serious complication, which has not previously been reported. The work described has not been submitted elsewhere for publication, in whole or in part, and all the authors listed have approved the manuscript that is enclosed. The manuscript is submitted to be considered as an Case Report.

I am the corresponding author and my address and other informations are as follows,

Address: Department of Cardiac Surgery, West China Hospital, Sichuan University, No. 37, Guoxue Alley, Chengdu, Sichuan Province, 610041, China.
Tel.: +86 13699487826,
Fax: +86 28 85503817,
E-mail: donglikn199@163.com

Thank you very much for your attention and consideration.

Sincerely yours,

Yuqing Wang